Author’s response to reviews

Title: Management of aplastic anemia during pregnancy: a case report

Authors:

Krista J.M. Stibbe (kstibbe@hotmail.com)
Hajo I.J. Wildschut (h.wildschut@erasusmc.nl)
Pieterella Lugtenburg (p.lugtenburg@erasmusmc.nl)

Version: 4 Date: 18 August 2010

Author’s response to reviews:

Dear Editor,

Following your email of August 11 requesting informed consent by next of kin I send you the permission the patient gave us before she died together with the informed consent of her parents (see material files).

I mentioned this informed consent also in the Consent section of the manuscript.

With best wishes,

Krista Stibbe
Erasmus University Medical Center
Rotterdam, the Netherlands